Author's response to reviews

Title: Diagnostic criteria for diabetes revisited: Making use of combined criteria

Authors:

Ali Parappil (aliparappil@yahoo.com)
Suhail A.R. Doi (sardoi@onebox.com)
Kamal A.S. Al-Shoumer (kshoumer@hsc.kuniv.edu.kw)

Version: 6 Date: 15 Jan 2002

Dear Sir

This version being uploaded has three editing changes:

1. Author affiliation to include Kuwait University
2. Legend individually written for figures and tables
3. Reference to table 2 included at the appropriate place in discussion

Kind regards

Dr Doi